Author's response to reviews

Title: Comparison of Trunk and Spine Deformity in Adolescent Idiopathic Scoliosis

Authors:

Brandon B Carlson MD, MPH (bcarlson@kumc.edu)
Douglas C Burton MD (dburton@kumc.edu)
Marc A Asher MD (masher@kumc.edu)

Version: 2 Date: 29 October 2012

Author's response to reviews:

One numbering error was found in the "Inclusion Criteria" section. This was not previously identified during editing and is now corrected.